Author’s response to reviews

Title: ACUTE RESPIRATORY FAILURE SECONDARY TO A CERVICAL GOITRE IN A PREGNANT WOMAN: A CASE REPORT.

Authors:

sidi driss el jaouhari (sidijhr@gmail.com)
nawfal doghmi (eljaouharichama@gmail.com)
hamza najout (chamitta05@gmail.com)
Massine El hamouni (drsaarahelabbassi@gmail.com)
El hassane Kabiri (driss200700@hotmail.com)
Hicham Bekkali (dr_elabbassi_86@hotmail.fr)
Jaafar Salim Lalaoui (nevermin_617@hotmail.com)
mustapha bensghir (faouziaguedira@gmail.com)

Version: 2 Date: 22 Jan 2019

Author’s response to reviews:

Response to the comments

ACUTE RESPIRATORY FAILURE SECONDARY TO A CERVICAL GOITRE IN A PREGNANT WOMAN: A CASE REPORT.

Doctor Sidi Driss El jaouhari

Thank you very much, Reviewers and Editor-in-Chief to have read and correct this article.

- Please ensure that author 'Massine El hammouni' has been spelled correctly within your manuscript and on the submission system- currently they differ.
- The name of Dr. el hamouni massine has been rectified

- Please move your Keywords section to below your abstract.

- The requested changes are made.

(PAGE 2)

- Please insert the heading 'Declarations' for the Declarations section, above the 'Abbreviations' section of your manuscript.

- The requested changes are made.

(PAGE 8-9)

- Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship). Currently, the contributions of authors SL, HN & HB do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

- SDE, ND and MB analyzed and interpreted the patient data. SDE and MB were major contributions in writing the manuscript. ME and EK performed the surgical gesture. JSL, HN, HB and MB interpreted the data, conducted a literature review and the final correction. All authors read and approved the final manuscript.

The role of different authors has been clarified.
(Author’s contributions, page 8).

- Please ensure to uploaded your figures with your resubmissions. Figures should be provided as separate files, and each figure of a manuscript should be submitted as a single file.

☐ Request made

- Thank you for providing a response to the reviewers. As this document is no longer required at this stage of the publication process, please remove it from your submission’s supplementary files.

☐ Request made

- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

☐ Request made